Author's response to reviews

Title: Effects of low-flow sevoflurane anesthesia on renal function in low birth weight infants

Authors:

Na Xing Dr (mzknana@126.com)
Xin Wei Dr (787611423@qq.com)
Yanzi Chang Dr (Minerva@163.com)
Yingying Du Dr (duyy87@126.com)
Wei Zhang Dr (15003995323@163.com)

Version: 3
Date: 3 May 2014

Author's response to reviews: see over
Dear Editor,

We submit the attached manuscript entitled “Effects of low-flow sevoflurane anesthesia on renal function in low birth weight infants” to be considered for publication in the BMC Anesthesiology.

In this study, we investigated the effects of low-flow sevoflurane anesthesia on renal function and clinical safety in low birth weight infants. Our study indicated that low-flow sevoflurane anesthesia did not significantly influence renal function in low birth weight infants undergoing abdominal surgery.

All authors have reviewed the final version of the manuscript and approve it for publication. This manuscript has not been published in whole or in part nor is it being considered for publication elsewhere. The authors have no conflicts of interest to declare. In advance of submission, we had this manuscript copyedited by an English speaking professional with science background at Elixigen Corporation (Huntington Beach, California, USA).

Thank you very much for your considering our manuscript for potential publication. I'm looking forward to hearing from you soon.

Sincerely,

Na Xing